Course Information Sheet, CMPUT204, Fall 2007, Section A2/EA2

This sheet contains basic information you need to know about this course. For more details see the course web page:

http://www.cs.ualberta.ca/~mreza/courses/204-F07/

Note: All announcements, assignments, and lecture notes will be posted on this web page. It is your responsibility to check this web page frequently (at least twice a week). Instructor Information:

• Instructor: Mohammad R. Salavatipour • E-mail: mreza@cs.ualberta.ca • Office: Ath303 • Phone: 492-1759

The best way to contact me is by e-mail. I usually reply within 1-2 days, but if you have questions about an assignment it is better to ask early.

Time tables:

• Lectures:

– Section A2/EA2: MWF 12:00-12:50, CAB 269. • Tutorials (seminars):

– Mon, 5:00-5:50, CAB357 – Tue, 12:30-1:20, CAB229 – Wed, 11:00-11:50 CAB377 • Office hours: Fridays 2-3.

Text book:

The text book for this course is: “Introduction to Algorithms (second edition)” by T. H. Cormen, C. E. Leiserson, R. L. Rivest, and C. Stein. McGraw Hill. 2001.

Additional reading: “Algorithm Design” by J. Kleinberg and E. Tardos. Addision Wesley, 2005.

Grading Scheme:

• Five assignments: 35% (5 at 7% each).

• Two term tests in class (50 minutes each) 30% (15% each). • Final exam: 35%.

Important Dates:

Date Event Wed, Sep 5 First lecture (Assignment 1 is posted) Wed, Sep 26 Assignment 1 due (Assignment 2 is posted) Mon, Oct 8 Thanksgiving (no lecture) Wed, Oct 10 Assignment 2 due Wed, Oct 17 Term test 1 (Assignment 3 is posted) Wed, Oct 31 Assignment 3 due (Assignment 4 posted) Fri, Nov 12 Remembrance day (no lecture) Wed, Nov 14 Assignment 4 due Wed, Nov 21 Term test 2 (Assignment 5 posted) Wed, Dec 5 Assignment 5 due, Last lecture

Assignments, late policy, and Plagiarism:

All assignments are due by noon (on the due date). Late assignments are not accepted. All cases of plagiarism will be forwarded by the instructor to the Dean. For more information see the course web page and also follow the link to The Code of Student Behavior.